Author’s response to reviews

Title: Assessment of intracranial pressure with ultrasonographic retrobulbar optic nerve sheath diameter measurement

Authors:

Dachuan Liu (ldc99@sina.com)
Zhen Li (zhenli5188@hotmail.com)
Xuxiang Zhang (zhangxuxiang@vip.163.com)
Liping Zhao (547055672@qq.com)
Jianping Jia (jiajp@vip.126.com)
Fei Sun (sf69@163.com)
Yaxing Wang (narvita@gmail.com)
Daqing Ma (d.ma@imperial.uk)
Wenbin Wei (weiwenbintr@163.com)

Version: 4 Date: 28 Aug 2017

Author’s response to reviews:

Dear Editor,

We are glad to receive your letter. We put the Figure as requested changes and upload again Data check method described in the following.

Enter http://www.medresman.org/ web page, click on the public access. In the upper right-hand corner to choose language - English. In the Trial search options, click more options, the study leader box input “dachuan”, and then click search, this study of the subject (Assessment of intracranial pressure with ultrasonographic retrobulbar optic nerve sheath diameter measurement) displayed on the Public title list.
Best wishes!

Yours sincerely,

Dachuan Liu

On behalf of all authorship.